

Article



# UAV Based Spatiotemporal Analysis of the 2019–2020 New South Wales Bushfires

Fahim Ullah 1,\*, Sara Imran Khan <sup>2</sup>, Hafiz Suliman Munawar <sup>3</sup>, Zakria Qadir <sup>4</sup> and Siddra Qayyum <sup>3</sup>

- <sup>1</sup> School of Civil Engineering and Surveying, University of Southern Queensland, Springfield, QLD 4300, Australia
- <sup>2</sup> Faculty of Chemical Energy, University of New South Wales, Sydney, NSW 2052, Australia; saraimrankhan17@gmail.com
- <sup>3</sup> School of Built Environment, University of New South Wales, Sydney, NSW 2052, Australia; h.munawar@unsw.edu.au (H.S.M.); s.qayyum@unsw.edu.au (S.Q.)
- <sup>4</sup> School of Computing Engineering and Mathematics, Western Sydney University, Locked Bag 1797, Penrith, NSW 2751, Australia; z.qadir@westernsydney.edu.au
- \* Correspondence: fahim.ullah@usq.edu.au

Abstract: Bushfires have been a key concern for countries such as Australia for a long time. These must be mitigated to eradicate the associated harmful effects on the climate and to have a sustainable and healthy environment for wildlife. The current study investigates the 2019-2020 bushfires in New South Wales (NSW) Australia. The bush fires are mapped using Geographical Information Systems (GIS) and remote sensing, the hotpots are monitored, and damage is assessed. Further, an Unmanned Aerial Vehicles (UAV)-based bushfire mitigation framework is presented where the bushfires can be mapped and monitored instantly using UAV swarms. For the GIS and remote sensing, datasets of the Australian Bureau of Meteorology and VIIRS fire data products are used, whereas the paths of UAVs are optimized using the Particle Swarm Optimization (PSO) algorithm. The mapping results of 2019–2020 NSW bushfires show that 50% of the national parks of NSW were impacted by the fires, resulting in damage to 2.5 million hectares of land. The fires are highly clustered towards the north and southeastern cities of NSW and its border region with Victoria. The hotspots are in the Deua, Kosciu Sako, Wollemi, and Yengo National Parks. The current study is the first step towards addressing a key issue of bushfire disasters, in the Australian context, that can be adopted by its Rural Fire Service (RFS), before the next fire season, to instantly map, assess, and subsequently mitigate the bushfire disasters. This will help move towards a smart and sustainable environment.

**Keywords:** bushfires; disaster management; spatiotemporal analysis; unmanned aerial vehicles; UAV path planning; geographical information systems; New South Wales Australia

# 1. Introduction and Background

Disasters around the globe have been impacting the global economies since the dawn of time. These disasters include earthquakes, floods, hurricanes, tornados, landslides, tsunamis, bushfires, and others [1]. Disasters result in the loss of lives, properties, real estate, and livestock, with serious consequences for the economic development of the affected countries. Accordingly, critical infrastructure, communications, real estate, vegetation, and forests are lost, in addition to the lives that hinder regional development. The reason behind frequent and recurring fire seasons is attributed to climate change [2–4]. Climate change, deforestation, growing urban development, and utilization of combustible sources are increasing the global temperatures in countries around the world. This gives rise to bushfires and extreme weather [5]. Consequently, the global fire seasons are getting prolonged, and the daily temperatures are rising, which are predicted to worsen and be

Citation: Ullah, F.; Khan, S.I.; Munawar, H.S.; Qadir, Z.; Qayyum, S. UAV Based Spatiotemporal Analysis of the 2019–2020 New South Wales Bushfires. *Sustainability* **2021**, *13*, 10207. https://doi.org/10.3390/ su131810207

Academic Editors: Marc A. Rosen and Eben Broadbent

Received: 14 June 2021 Accepted: 10 September 2021 Published: 13 September 2021

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



**Copyright:** © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/). more severe if climate change issues are not addressed [6,7]. In the era of demands for global sustainability, it is imperative to address such climate issues.

Globally, bushfires and wildfires burn approx. 4% of the global land surface each year, which amounts to approx. 30–46 million km<sup>2</sup> of the global land surface [8]. However, due to its limited direct impact on individuals, it does not attract wider attention from the media that is sometimes more focused on reporting only the tragic impacts of such fires that directly impact human lives. As a result, these fires sometimes go unnoticed by the world and are only discussed and addressed in the affected country. In the last two decades, extreme bushfire incidents have been observed worldwide, causing immense economic, social, and environmental loss. The trend of these fires is increasing, and in recent years, this natural disaster has been experienced in the regions where the fires are an unusual event. These include Brazil, Bolivia, Chile, Sweden, and the Arctic Circle [9,10].

In Australia, bushfires are a frequent phenomenon due to their geographic location, varying temperatures, and other natural causes [5]. The deadliest fires in Australian history occurred in 1851, 1939, 1983, 2009, and recently in 2019–2020. As per the report of conversation.com.au, the costs of bushfires have passed \$100 billion in Australia, making it the costliest natural disaster [11]. Australia's 2019–2020 bushfire season, known as the Black Summer Fires, is regarded as one of the worst bushfire events in the country's history that had serious impacts on wildlife, forests, agricultural land, and public properties. It impacted infrastructure in the states of New South Wales (NSW), Victoria, Queensland, and others. The Black Saturday fires alone burnt 430,000 hectares of land. Overall, the Black Summer fires burnt 10.7 million hectares, equivalent to an area of the size of South Korea or Scotland and Wales combined.

Further, more than a billion wildlife animals have been killed, and over 2000 homes were destroyed due to these bushfires [12]. The major causes of the 2019–2020 bushfires included the extended drought conditions, which left dry bushlands and forests that acted as extremely dry and spatially contiguous fuel spreading through forested regions of NSW stretching from Queensland to Victoria. Several large-scale climate drivers contributed to this dryness of 2019 summer, including a strong and long-lived positive Indian Ocean Dipole and negative Southern Annular Mode. Further, the dryness of the landscape was also influenced by reduced cool-season rainfall and other long-term climate trends. Due to extreme fires, it was challenging to quickly detect and extinguish new ignitions in remote areas where they started, resulting in delayed responses that fuelled the intensity of existing fires and strained the resources. Further, the intensified and dense smoke, due to multiple fire events, made it impossible to know where the fire edge was with precision because line scanner aircraft could not fly, and alternate infra-red scanning was a low resolution or unavailable. These issues made the firefighting more difficult, and the fires grew out of control. The limited capacity to fight fires at night led to many fires taking big runs at night and early mornings, causing havoc in the Australian states of NSW, Victoria, and their border regions [13].

NSW is selected as the study area due to its history of bushfire events. It experiences frequent fires due to its widespread vegetation and bushes that fuel draughts and extreme temperatures. Table 1 shows some of the impactful fire seasons in NSW since 1965. In 1965, NSW observed 251,000 hectares of damage, damaging 59 homes, and causing three fatalities [14]. In the 1984–85 fire season, much of the damage was experienced in the loss of livestock and \$40 million economic loss [15]. Similarly, in 2013, the Warrumbungle fires impacted 53 houses, 118 buildings, and damaged agricultural infrastructure and buildings. The 2017 Wentworth falls winter fires damaged an area of 52,000 hectares and destroyed 35 homes. The statistics signify that, though the human loss due to these fires in the past had been low, the property and economic loss had been severe in each fire event. Further, in the growing era of sustainability and wildlife protections, it is imperative to devise plans that reduce the impacts of such fires on global climate, wildlife, and public properties. Nevertheless, the latest fires associated with the Black Summer event had been tragic in all aspects for the state of NSW and Australia in general. These have resulted in

Sr.	Year	<b>Fire Events</b>	Area (Hectares)	Deaths	Property Damage	Sources
1.	1965	Southern Highlands Bushfire	251,000	3	59 homes	[14]
3.	1980	Waterfall Bushfires	1,000,000	5	14 homes	[16]
4.	1984– 1985	NSW Fire season	3,500,000	5	0 homes, 40,000 livestock, and \$40 million economic damage	[15]
5.	1994	Eastern seaboard fires	400,000	4	225 homes	[17]
6.	2002	Black Christmas bush- fires	753,314	0	121 homes	[18]
7.	2006	Jail Break Inn Fire, Junee	30,000	0	7 homes, 4 non-residential build- ings	[19]
8.	2006	Pulletop bushfire, Wagga Wagga	9000	0	2 homes, 2500 livestock, 3 vehicles, 50 km of fencing	[20]
9.	2006– 2007	Australian Bushfire sea- son	1,360,000	8	83 homes, 20 non-residential build- ings	[21]
10.	2013	Warrumbungle	54,000	0	53 homes, 118 other buildings, Ag- ricultural equipment, and infra- structure at Siding Spring Observa- tory	[22]
11.	2013	2013-NSW fires	100,000	1	208 homes, 40 non-residential homes	[23]
12.	2017	Wentworth falls Winter Fire	52,000	0	35 homes	[24]
13.	2019– 2020	Black Summer	1,700,000	33	2439 homes, more than billion wild animals lost	[13]

 Table 1. Major bushfires in New South Wales from 1965 to 2020.

and more than a billion wild animals lost [13].

the loss of 33 lives, 10.7 million hectares of land burnt, more than 2400 homes destroyed,

The state of NSW used several remote sensing techniques during the 2019–2020 fire seasons to assess the fire damages. Fire and land management agencies at state and federal levels have remote sensing capabilities that provided useful information during the planning, preparation, and response phases of the 2019–2020 bushfire season. NSW rural fire service (RFS) uses remote sensing technologies in various ways. In a report, NSW RFS reported that its firefighters on the ground and in vehicles provided the best intelligence they could on fires, considering the extent and scale of the fires. Further, it found camera platforms on helicopters with infra-red and high-definition imagery useful. Further remote sensing data from multispectral scanning devices ('line scanners') mounted on contracted fixed-wing aircraft was particularly helpful in assessing bushfire movements, spread, and damages [25]. The NSW RFS reports that, across 165 days during the 2019–2020 season, a total of 565-line scanning flights were flown, amounting to 7469 flight hours.

Another high potential solution to addressing the Australian bushfires emergencies is UAV usage that does not rely on human pilots and has very little potential for data losses. UAVs have been used in various fields such as smart cities, real estate, property management, healthcare, construction, agriculture, and others [26–32]. These have been extensively explored in addressing disaster situations such as emergency evacuation path planning, flood response, and others [33–36].

Given that they do not require the presence of human resources, such vehicles can be readily made available to use in case of emergencies such as bushfires. UAVs are thus ideal tools that have been used in a post-disaster scenario. For example, in March 2015, when Tropical Cyclone Pam, a category 5 storm, struck Vanuatu, UAVs were employed for mapping areas to assess the damage [37]. Similarly, when Typhoon Haiyan affected Philippines in November 2013, UAVs were used for aerial imaging to get real-time information on the disaster with the help of aerial imaging and the development of future frameworks for emergency response planning [38]. In a recent study, these have been proposed to be used for bushfires in Victoria Australia [5].

Owing to the nature of the frequently occurring bushfires in Australia and NSW, the current study aims to develop a novel framework for assessing the bushfires and applies UAV swarm knowledge for better decision-making and pertinent hazard mitigation strategies. It uses a geographical information system (GIS) based assessments of the fire hot spots of NSW. Further, an unmanned aerial vehicle (UAV) based framework is presented to monitor the bushfire hotspots and instigate immediate rescue measures for minimizing the impacts of these bushfires. The paths of the UAVs are optimized for effective monitoring of the affected areas with the potential to deliver any information, ration, or first aid kits. Specifically, the PSO model, available at https://www.mathworks.com/matlabcentral/fileexchange/69027-simulation-of-particles-in-particle-swarm-optimization (accesed 9 August 2021) has been used in the study as a base model that has been considerably modified and expanded to suit the needs of the current study.

The rest of the paper is organized as follows. Section 2 presents the potential tools and techniques to help bushfires management. These include the GIS and remote sensing tools and UAVs. Section 3 presents a comprehensive overview of the methodology of the current study and lists key assumptions, constraints, and model codes used in the study for UAV routing problems. Section 4 discusses the key results of the GIS-based assessment of the fire hotspots and the routing results for UAV path planning. Finally, Section 5 concludes the study and presents the future directions to build upon the current study.

#### 2. Potential Tools and Techniques for Bushfires Management

Satellite data obtained through GIS and remote sensing is a widely used primary source of information for the active mapping of fire and burned areas at regional to global scales. The Moderate-resolution Imaging Spectroradiometer (MODIS) from NASA Terra and Aqua satellites were the first satellite-borne sensors with the ability of monitoring fire radiative energy (FRE) release rate, or power (FRP), quantitatively on a worldwide scale. Researchers around the globe have used these to assess wildfires [39,40]. Two kinds of satellite data are used to detect fire events: active fire and burned area products [41]. Burned area products are based on the variations in the reflectance or with the combination of reflectance and active fires [42,43]. In comparison, active fire products are dependent on the detection of thermal anomalies [44].

The GIS tools that enable the monitoring of bush fire hotspots are kernel density and Getis-Ord Gi\* hotspot analysis. The kernel is a widely used estimator that helps generalize or smoothen discrete point data into a continuous surface area [45]. On the contrary, Getis-Ord uses the Gi\* statistics to calculate the degree of correlation of weighted features in the specific distance threshold. It can be used to identify the clustering pattern in the study area [46,47]. Gi\* statistics benefit from concurrently capturing the frequency of the events, the corresponding values, and the spatial relationship [46]. These simple yet efficient GIS tools have extensive applications as spatial, such as and cluster analysis tools in bushfire hazard management. Accordingly, remotely sensed data coupled with GIS tools could facilitate the local administrations to reduce natural disasters such as bushfires [42].

In NSW, remote sensing is an invaluable aid in predicting the weather, climate and assessing fire location. It has been used to assess fire conditions, extent, and behaviour by the NSW Rural Fire Service (RFS) in the 2019–2020 bushfire season. However, Australia's capabilities in this field have not been harnessed to fight bushfires swiftly and properly. Currently, these tools are only used after a fire is initiated due to a lack of automation and implementation in Australian contexts. Remote sensing must be properly adopted for automatic sensing of fire for big fire-risk seasons. The positive points are that Australia has

developed infrastructure and defence agencies that are already using remote sensing techniques, which can be shared with other departments for bushfire management. The key improvement areas include the enhanced capability for early detection of new ignitions, real-time tracking of the fire edge progression and intensity as it spreads, and a better understanding of vegetation and fuel load issues before the fires start. Such remote sensing can monitor and analyse the causal factors of a bush fire, inform the planning department to prepare for, and promptly respond to a bushfire event. Accordingly, a spatial technology acceleration program is needed in NSW (and Australia) to maximize the information available from all the various remote sensing technologies currently used.

For 214 days, from 10 August 2019 to 11 March 2020, NSW RFS flew its line scanning aircraft for at least some time on 88% of days. These line scanning techniques produce good quality imagery above active bush fires, making it possible to see details of the fire edge, its extent, and intensity. Such imagery helped NSW RFS make informed decisions about resource commitments and public warnings during fire events at the height of the 2019–2020 bushfire season. Not downplaying the efforts of the NSW RFS, the destruction levels are still well above the acceptable levels, and the techniques must be revised and improved to minimize the level of destruction. Accordingly, more line scanning and remote sensing techniques can help improve disaster response. Given the relatively low number of aircraft available, and the number of large fires raging simultaneously, only a relatively small number of line scanner 'snapshots' of each fire had been possible in the Black Summer. This is a serious drawback given the highly dynamic and dangerous nature of these fires. Further, a drawback of any sensor mounted on piloted, fixed-wing aircraft is that the sensor is useless when the plane cannot fly, as smoke/dust/fog makes flying impossible. The NSW RFS estimates that there were 26 days between 10 August 2019 and 11 March 2020 when line scanning aircraft could not be used at all due to ambient conditions affecting visibility or resourcing considerations. It is important to note that these figures do not include instances where scanned imagery was insufficient or where scans could not be completed frequently. While this is a relatively small period (12%), this inability to fly can be an issue when information about new ignitions, edges, and spread is needed instantly. Thus, it is imperative to explore new techniques and improve the existing ones for better bushfire management. A candidate for this is the usage of a UAV.

UAVs have been used in bushfires assessment, in bushfire hotspot detection [48], and economic evaluation of wildfires through UAVs [49]. Similarly, conceptual discussions have been retrieved from literature as relevant to fire monitoring with UAVs through cognitive human-machine interfaces and interactions [50], as well as remote sensing, to assess grapevine canopy damage due to fire smoke [51] and improve readiness for the next major bushfire emergency [52]. In the case of Australia, these have been proposed to assess bushfires in the state of Victoria [5].

Different types of algorithms exist for planning and optimizing UAV paths efficiently and effectively to reduce associated costs. These include Java-based algorithms, such as greedy, inter-route, intra-route, Tabu, and particle swarm optimization (PSO) [53,54]. Other studies have used Ant Colony Optimization (ACO) and Genetic Algorithm (GA) techniques. However, PSO has never been used for path planning of UAVs for bushfire assessment in NSW Australia, which is a humble contribution of the current study. PSO is a heuristic method that starts its search process using an initial particle population [55– 57]. Each particle represents a potential solution to the problem [58]. There is a multidimensional search space where these particles move around until they reach a constant state or the computational constraints are fully exhausted. PSO mimics the behaviour of birds in a flock or sheep in a herd [59]. It is based on a collection of particles in a swarm where each particle represents a possible solution to the problem. Due to its established advantages, PSO has been utilized in the current study, taking advantage of its ease of implementation, few parameters to adjust, robust, higher efficiency in finding the global optima, converge quicker, short computational time, and no overlapping. A comprehensive, optimized UAV system has not been proposed, to date, in the relevant literature for assessing bushfires issues and subsequent management in the Australian context. This gap is targeted in the current study, where the applications of UAVs are proposed and tested in the NSW region of Australia that is prone to frequent bushfires. Overall, the current study uses a mix of remote sensing and GIS for bushfire hotspot assessments and develops a UAV based optimized path system for instigating a swift emergency response to help mitigate bushfire disasters. Burnt locations and hotspots of 2019– 2020 bushfire season in the NSW are assessed using GIS tools and remote sensing data of VIIRS. Further, the statistical significance of these fire events, using the geostatistical tool of Getis Ord Gi\* statistics, is also assessed to discuss the impact of damage by the 2019– 2020 fires.

#### 3. Materials and Methods

This study follows a systematic approach for addressing the bushfires disasters in NSW regions of Australia. A four stepped method is adopted in the current study, as shown in Figure 1. In the first step, a review of the data available about global and Australian bushfires is conducted, as evident from the introduction section of the current study. This is augmented with the data about fires in NSW. Afterward, GIS, remote sensing applications, and UAVs for bushfire assessments are discussed in the section of tools and techniques available for bushfire assessment. In the second step of the current study, GIS-based assessments and burnt area monitoring are performed using data from VIIRS, and a UAV-based bushfire assessment framework is presented. In step three, the paths of UAV swarms are optimized using the PSO algorithm to identify the shortest possible paths for covering the bushfires area. In the fourth and final step of the study, the GIS-based bushfire monitoring reports are presented along with the regression analysis for bushfires-related socio-economic loss assessments. Lastly, the PSO-based UAV optimization results are presented to discuss the best routes for UAVs in mitigating bushfires disasters and instigating a swift response.



Figure 1. Methodology of the current study.

# 3.1. Study Area

The case study area of the current study is the state of NSW, Australia. The state is in the southeast of Australia on the eastern coast. It houses Sydney, the most populated city in Australia, and is among the top revenue-generating states in Australia. The area of NSW is 801,150 km<sup>2</sup> and has a population of approx. 8.092 million as of 2020. Vegetation bushlands cover over 80% of NSW and forests, as shown in Figure 2a, making it a frequent bushfire experiencing state. The motivation to choose the NSW for this study is that the state was set ablaze in the recent Black Summer fires and has reported the highest loss among all Australian states. The state observed 10,520 fire incidents in its various parts, destroying 75% of the total infrastructure losses of the Black Summer fires. Figure 2b provides the total fire events for NSW, based on GIS and remoting sensing, using the data from VIIRS sensors.



(a) NSW Map with Vegetation

(b) NSW Fire Events

Figure 2. New South Wales map with total fire events from the VIIRS sensors.

#### 3.2. Study Datasets

Table 2 summarizes the dataset used in the study. The primary dataset used for this study is the National Oceanic and Atmospheric Administration (NOAA) data of the VIIRS fire product. The VIIRS empowers operational environmental monitoring and numerical weather forecasting. It has 22 imaging and radiometric bands covering wavelengths from 0.41 to 12.5 microns. It provides sensor data records for more than twenty environmental data records such as the clouds, sea surface temperature, ocean colour, polar wind, vegetation fraction, aerosol, fire, snow and ice, vegetation, etc. The on-orbit verification in the postlaunch check-out and intensive calibration and validation have shown that VIIRS is performing very well. It has been used in the current study due to its precise resolution of detecting the smallest fires. The VIIRS sensor identified 10,446 fires, including minor and major fires, within NSW for the 2019–2020 Black Summer time. Apart from the fire product, climatic data for mean temperature, mean rainfall, and Forest Fire Danger Index (FFDI) are acquired from the Australian Government Bureau of Meteorology. This data is used to study the underlying climatic conditions responsible for these bushfires. These datasets monitor the burnt area and bushfire hotspots in the NSW region for the Black Summer period.

Table 2. The datasets used for the stuc	٩y.
---	-----

Data	Туре	Date of Acquisition	Source
Fire points	VIIRS	14 October 2020	Fire Information for Resource Management Sys- tem (FIRMS)
Temperature	Mean Temperature Deciles	14 October 2020	Australian Government Bureau of Meteorology
Rainfall	<b>Rainfall Deciles</b>	14 October 2020	Australian Government Bureau of Meteorology
Forest Fire Danger In- dex (FFDI)	FFDI Deciles	14 October 2020	Australian Government Bureau of Meteorology

### 3.3. GIS Analyses of the NSW Bushfires

Figure 3 provides the methodology flowchart to monitor the burnt area and map bushfire hotspots in the study area. The data and annual reports, from the Australian Bureau of Meteorology, of the region of interest were acquired. An in-depth review and assessment were used to relate and understand the fire patterns and areas identified in the



analysis. Accordingly, four key steps were performed: burned area mapping, fire clustering, hotspot monitoring, and environmental conditions and impact.

Figure 3. Methodology flowchart for mapping the bushfires hotspots in NSW.

For mapping the burnt area, ArcGIS software is used in which the interpolated perimeters from the monthly accumulated fire points are generated using a convex hull aggregation with the 'aggregate points' tool. The convex hull algorithm assigns an area including the clusters of points (minimum 3) at user-defined aggregation distance. Three aggregation distances, 1 km, 2 km, and 5 km, are tested for the fire delineation. These distances are chosen depending on the spatial resolution of the active fire products from the VIIRS-375 m resolution. The idea is to visualize the total burned area due to the fires in the NSW region. The validation of the fire samples is performed using visual interpretation from Google Earth imagery. The High/Low Clustering (Getis-Ord General G) provides the fires' gathering pattern that is used to measure the extent of clustering in the fire data.

The z-score and *p*-value depict the statistical significance of the null hypothesis. In this case, the null hypothesis states that the values linked with each feature are distributed randomly. For monitoring the hotspots, Getis-Ord local Gi\* spatial statistics is performed to see the statistical significance of the fire incidents. Before the incremental spatial autocorrelation tool is operated, beginning distance and distance increment must be set. Calculate Distance Band from the Neighbor Count tool is used to monitor these parameters. The tool gives the minimum, average, and maximum distance at which each point has at least one neighbour. The resultant maximum distance is used as the beginning distance, whereas the average distance achieved from the tool is used as the distance increment. Later, the incremental spatial autocorrelation tool is used to measure data grouping in space. The tool gives an output in the form of a graph of increasing distances and their corresponding z scores.

The clustering distance is subsequently used in the Getis-Ord  $Gi^*$  analysis as a distance band or radius. The Getis-Ord local statistic is calculated using Equations (1)–(3).

$$Gi^{*} = \frac{\sum_{j=1}^{n} wij.xj - \bar{x} \sum_{j=1}^{n} wij}{S \sqrt{\frac{\left[n \sum_{j=1}^{n} wi^{2}.j - \sum_{j=1}^{n} wij\right]^{2}\right]}{n-1}}}$$
(1)

$$\bar{x} = \frac{\sum_{j=1}^{n} x_j}{n} \tag{2}$$

$$S = \sqrt[2]{\left[n\sum_{j=1}^{n} x. j^{2} - (\bar{x})^{2}\right]}$$
(3)

where xj is the attribute value for feature j, wi.j is the spatial weight between the feature iand j, and n is the number of features.  $\bar{x}$  is the mean of all measurements, and S is the standard deviation of all measurements. The  $Gi^*$  is a zone, after which no more calculations are required. The  $Gi^*$  statistic returned for the features in the fire datasets is a z-score. For the z-scores to be statistically significant, the higher the z-score value, the more intense the cluster will be, hence classifying it as a hot spot. Consequently, the cluster will have low values for statistically strong negative values, identifying it as a cold spot. Thus, the spots can be classified into hotspots or cold spots for assessing the fires. Lastly, a linear regression analysis is performed, with the response variables of the burned area, fire incidents, fatalities, and the predicting variable as the fire season (year). Positive and negative relationships are represented as increasing and decreasing trends, respectively.

### 3.4. PSO for UAVs Path Planning in Bushfires Monitoring

PSO is a metaheuristic algorithm that works on the principle of finding, generating, and searching for the shortest path. In this study, the PSO algorithm is used to monitor the bushfire area. It is quite challenging, in bushfires, to reach the allocated area and hover back to the depot. In the relevant literature, compared to the existing algorithms such as ACO and GA, PSO is favoured to generate the shortest distance with enhanced collision avoiding capability [60,61]. Moreover, it is the best possible approach to significantly find the shortest distance in optimum time [62,63]. In the current study, the PSO optimization algorithm used in UAV Bushfire Application is inspired by "Seyedali Mirjalili (2021). Simulation of particles in Particle Swarm Optimization", available at (https://www.mathworks.com/matlabcentral/fileexchange/69027-simulation-of-particles-in-particle-swarmoptimization, accessed on 9 August 2021). The document was accessed on 9 August 2021. Significant changes have been made to the source code, including changing the parameters, e.g., handle points, maximum iterations, population size, inertial weight, as well as personal and global learning coefficients that have been modified to be used in our specific application where the particles are the UAVs in a swarm. Specifically, three-point handles have been used in the current study compared to the source code. The function used in the current study is for five obstacles with different positions and diameters, which are some of the novel additions and modifications, of the current study, to the source code. Furthermore, based on the computation time and transportation cost, the current study considers the best path for UAVs to reach the affected area in the least time. For this purpose, the number of obstacles has been increased to five to make a complex environment for UAVs to reach the destination.

The flow chart in Figure 4 shows the simulation model for the PSO algorithm. Initially, the start and target points are determined for UAVs to fly from the depot (start) and reach the target location. In step 2, the population size and parameters are set for particle velocity, time steps, and personal and global learning coefficient. In step 3, the PSO algorithm model is run to maximize the affected area coverage based on the cost function generated in step 4. As a result, the random paths are generated, and the shortest path is selected based on the maximum iterations.



Figure 4. PSO simulation model.

Figure 5 provides a five-stepped framework for bushfire detection using UAVs through the PSO algorithm. In step 1, the control unit is notified regarding the affected region where the bushfire is ignited using field and satellite sensors. In step two, the control unit/van is sent to the nearest safe area of the bushfire. This is done to avoid unnecessary battery losses of the UAVs due to the hovering of UAVs as they have limited battery

power. It further ensures high endurance and better wireless communication with the UAV due to the shorter distance. In step 3, the GPS coordinates of the UAVs are set, and the required data is embedded to cover the targeted location for bushfire damage detection. In step 4, the PSO algorithm is initialized to determine the shortest path between the start point and the end destination, retrieving the data in lesser time and minimum transportation distance. Based on the shortest path calculated, the UAV swarm assigns the task to each other to minimize the energy consumption and better monitoring time. In the final step, the real-time fire is monitored using cameras and sensors attached to the UAV. The data is shared with the control unit in real-time, where it can be shared with all concerned departments. A rescue relief team is notified instantly to reduce the effects of bushfires.



Figure 5. Schematic illustration of Bushfire Detection using PSO.

Figure 6 illustrates the pseudocode and demonstrates the trajectory of UAVs from determining the maximum area coverage. In Figure 6, it is assumed that the B1, B2 are the Barriers, UAV1, UAV2, and UAV3 are the UAVs, while P1–P10 are the locations to be covered by the UAVs. The first step involves determining the target location, which is, identifying the fire zone. The UAVs initialize themselves from P1 to compute the feasibility paths in Step 2. These UAVs are launched from the control units or vans present in the vicinity of the fire zone. To minimize the transportation distance, the UAVs communicate with neighbouring UAVs and determine the shortest path possible in Step 3. In Step 4, the optimization method is adopted, where three of the PSO algorithm functions are run to maximize the area coverage, minimize the distance to the target, and minimize the number of active UAVs. As a result, an efficient and cost-effective disaster management strategy is devised whereby the UAVs can cover the maximum area in the shortest possible time, as given in Step 5, and where all the locations from P1 to P10 are covered by the three UAVs using PSO. This way, the barriers in the paths are avoided, as evident from



B1 and B2 in Figure 6, and an efficient maximum area coverage strategy is used to get data from the target zone.

Figure 6. Routing mechanism to calculate best transportation distance using PSO.

The area assignment for UAVs is based on the longitude and latitude coordinates using GPS technology [64,65]. Table 3 lists the parameters and assumed values of these parameters to assist the PSO algorithm. The inertia weight determines the contribution rate of a particle's previous velocity to its velocity at the current time step that is considered as 0.8 in the current study. The inertia weight damping ratio is assumed as 0.96. The personal learning coefficients and the global learning coefficients to fit the maximum area curve are 1.25. These values are taken from the studies of Mirjalili et al. [66] and Mirjalili et al. [67] based on the optimum results achieved.

Table 3. The PSO parameters and their values.

Parameter Values			
Inertia Weight			
Inertia Weight Damping Ratio			
Personal Learning Coefficient			
Global Learning Coefficient			

#### 3.5. Proposed Model

The parameters and functions of the proposed model are presented in Table 4.

Parameters	Functions
D	Set of depot sites
Т	Set of target sites
0, n + 1	The start and end depots of the drone
V	$V = D \cup T \cup (0) \cup (n+1)$
R	Set of UAVs
$q_i$	Requested demand for each target site $i \epsilon$ T
qr	Capacity of drone $r \in \mathbb{R}$
$\mathbf{P}_r$	Time required for UAV to cover area $r \in \mathbb{R}$
a h	Time window during which the UAV reaches the target
Ui. Di	site
$t_{ij}$	Time to travel from <i>i</i> to <i>j</i> , $j \in V$
r	Maximum time lag between consecutive deliveries

Table 4. Model parameters and functions.

The UAV starts its trip at a central (source) depot and travels between the depot and target sites. At the end of mission completion, the UAV returns to an end (sink) depot (which may or may not be the same as the starting depot). This routing problem can be modelled on a directed, weighted graph G (V, A); consisting of vertex set  $V = \{0\} \cup T \cup \{n+1\}$ , where vertices 0 and n + 1 are resp. to the source and sink depots. The UAV must reload in between 2 deliveries. This has been accounted for in the arc costs. Optionally, a positive load time for the UAVs can be added to the arcs between two target sites. The arc set A is defined as follows:

- The source-sink depots have outgoing resp. incoming edges to/ from all other vertices.
- There is an arc (i,j) for all  $i; j \in T; i \neq j$ .
- The arc costs are as follows:
- $T_{0,i} = \min_{p \in P} t_{i,p} + t_{p,j}$  for all  $i \in T$
- $T_{i,j} = \min_{p \in \mathbb{P}} t_{i,p} + t_{p,j} \text{ for all } i, j \in \mathbb{T}$
- $T_{i,n+1} = t_{i,n+1}.$
- $T_{0,n+1} = 0.$

The routing constraints can be modified as follows:

$$max.\sum_{i\in T}q_{iY_i} \tag{4}$$

$$\sum_{\epsilon R \in \delta + (i)} \sum_{i} qr XijR \ge qiYi \qquad \forall i \in T$$
(5)

Constraint (5) ensures that enough area is covered at the target site.

$$\sum_{j\in\delta+(0)} X0jr = \sum_{i\ \epsilon\ \delta(n+1)} Xi, n+1, r=1 \qquad \forall\ r\ \epsilon\ R$$
(6)

$$\sum_{j\in\delta+(i)} Xijr = \sum_{j\in\delta-(i)} Xj, i, r \qquad \forall i \in \mathbb{T}, r \in \mathbb{R}$$
<sup>(7)</sup>

Constraints (6) and (7) determine the shape of a feasible tour: a tour starts at the source depot, visits a target site, and finally returns to the sink depot. Constraints (7) are the flow preservation constraints. Further, between two consecutive visits, starting, processing, and travel times must be considered (Constraint (8)).

$$T_r^i + t_{ij} - M\left(1 - X_{ijr}\right) \le T_r^j - D_r \qquad \forall \, i, j \in V, r \in \mathbb{R}$$
(8)

$$a_{i+}D_r \le T_r^j \le b_i \qquad \forall i \in \mathbb{V}, r \in \mathbb{R}$$
(9)

A target  $i \in T$  cannot be visited before  $a_i$ , and the assigned task must be completed before  $b_i$  (Constraint (9)). The remaining Constraints (10)–(12) restrict the domains of the variables.

$$X_{ijr} \in \{0, 1\} \qquad \forall i, j \in \mathbb{V}, r \in \mathbb{R}$$
(10)

$$T_r^i \in \mathbb{Z} \ge 0 \qquad \qquad \forall i \in \mathbb{V}, r \in \mathbb{R}$$
(11)

$$Y_i \in \{0, 1\} \qquad \forall i \in \mathbb{T}$$
(12)

where,  $X_{ijr}$  is a binary variable, indicating whether UAV *r* travels from *i* to *j*. Integer variable  $T_r^i$ ,  $i \in T$ ,  $r \in R$ , record the time that UAV *r* finishes its delivery to target site *i*. For notation purposes,  $\delta^-(.)$  resp.  $\delta^+(.)$  denote the incoming resp. outgoing neighbourhood sets. The maximum time lag requirements between multiple tasks completed for a single site can be schedules as below:

$$max.\sum_{i \in T} q_{iY_i} \tag{13}$$

$$\sum_{r \in Rl \in R0 \setminus \{r\}} \sum_{i} q_r Z_{rl}^l \ge q_i Y_i \qquad \forall i \in T$$
(14)

Constraint (14) ensures that assigned tasks have been completed at each target site.

$$\sum_{i \in R0} Z_{r0l}^i = 1 \quad \forall \ i \in \mathbb{V}$$
(15)

Constraints (15) and (16) are used to sequence the UAVs. A UAV can only be used once for every target site, and whenever it is used, its delivery must be succeeded by another delivery (possibly a delivery by the dummy UAV *r*<sub>0</sub>) (Constraint (15)). The dummy UAV must be scheduled (Constraint (16)).

$$\sum_{l \in \mathbb{R} \setminus \{r\}} Z_{rl}^{i} = \sum_{l \in \mathbb{R} \setminus \{r\}} Z_{lr}^{i} \qquad \forall r, l \in \mathbb{R}, r \neq l, i \in \mathbb{V}$$
(16)

Constraint (17) is the flow preservation constraint. Together with Constraint (16), these constraints enforce that all assigned tasks are scheduled consecutively. Thus, each task has exactly one successor and one predecessor.

$$T_r^i - M\left(1 - Z_{rl}^i\right) \le T_{rl}^i - D_l \qquad \forall r, l \in \mathbb{R}, r \neq l, i \in \mathbb{V}$$

$$(17)$$

Constraint (17) links the completion time variables  $T_r^i$  and the sequence variables  $Z_{rl}^i$ , thereby enforcing that tasks performed do not overlap in time.

$$T_{l}^{i} - D_{l} \leq T_{r}^{i} + \gamma + M \left( 1 - Z_{rl}^{i} \right) \qquad \forall r, l \in \mathbb{R}, r \neq l, i \in \mathbb{V}$$

$$(18)$$

Constraint (18) enforces a maximum lag time between consecutive assigned tasks.

$$a_{i+}D_r \le T_r^J \le b_i \qquad \forall i \in V, r \in \mathbb{R}$$
<sup>(19)</sup>

Constraint (19) ensures that assigned tasks performed by the UAVs are scheduled within the time window. The remaining Constraints (20)–(22) restrict the domains of the variables.

$$Z_{rl}^{i} \in \{0, 1\} \qquad \forall i \in \mathbb{V}, r, l \in \mathbb{R}$$

$$(20)$$

$$T_r^i \in \mathbb{Z} \ge 0 \qquad \forall i \in \mathbb{V}, r \in \mathbb{R}$$
(21)

$$Y_i \in \{0, 1\} \qquad \forall i \in \mathbf{T}$$
(22)

The binary variable  $Z_{rl}^i$  is equal to one if UAV  $r \in \mathbb{R}$  completes its tasks immediately before UAV  $l \in \mathbb{R}$  to target site  $i \in T$ ; otherwise, the value is zero. In this model,  $r_0$ represents a dummy drone,  $\mathbb{R}_0 = \mathbb{R} \cup \{r_0\}$ . A feasible solution is obtained at the intersection of the routing and scheduling polytopes. Connecting the two polytopes is accomplished via the linking constraints:

$$\sum_{l \in RO \setminus \{r\}} Z_{rl}^{i} = \sum_{j \in N} X_{ijr} \qquad \forall r \in \mathbb{R}, \quad i \in \mathbb{T}$$
(23)

Note that Constraints (14) and (17) in the route are identical to Constraints (5) and (9) in the schedule, respectively, and are consequently dropped.

A disadvantage of the current model is that a single UAV cannot visit the same target site more than once. This restriction is unrealistic as, often, a UAV can travel back and forth between the depot and a target site. When the binary variables  $X_{ijr}$  are replaced by equivalent integer variables, indicating the number of times UAV R travels from *i* to *j*, one can still distinguish the routes. However, expressing the scheduling constraints becomes difficult in this case. Two options exist to address this issue: either the distinct trips made by a single UAV are enumerated (e.g., UAV R travels from *i* to *j* during trip *t*), or the visits to a target site are enumerated. The latter solution is applied in assignment-based formulations for scheduling problems. This model is adjusted to our notation as below:

Let D and T be defined as above. In addition, for each target site  $i \in T$ , a new ordered set consisting of visits to the target site,  $T^i = \{1, \dots, (i)\}$ , is defined where  $n(i) = [q_i/\min_{r \in R} (qr)]$  is an upper bound of visits required by the target site i. A shorthand notation,  $t_j^i$  will be used to denote visit j for target i. A time window  $[a_e, b_e]$  is associated with each visit  $e \in T^i$ ,  $i \in T$  which is initialized to the time window for the corresponding site  $i \in T$  i.e.,  $[a_e, b_e] = [a_i, b_i]$  for all  $i \in T$ ,  $e \in T_i$ . Finally,  $W = U_{i \in T}T_i$  is the combination of all the visits.

Let directed weighted graph be G (V, A); consisting of vertex set V=  $\{0\} \cup W \cup \{n + 1\}$ . Its arc set is defined as follows:

- The source-sink depots have outgoing resp. incoming edges to/from all other vertices
- A delivery/trip node  $t_h^i$  has a directed edge to a trip node  $t_j^i$  if  $h < j, i \in T, h, j \in T_i$ .
- There is a directed sedge from  $t_u^i$  to  $t_{v'}^j$ ,  $i \neq j$ , except if  $t_v^j$  needs to be scheduled earlier than  $t_u^i$ .

The arc costs are as follows:

- $T_0, t_j^i = \min_{p \in \mathbb{P}} tm_{0,p} + tm_{p,i} \text{ for all } t_j^i \in \mathbb{W}.$
- $T t_u^i t_v^j = \min_{p \in \mathbb{P}} tm_{i,p} + tm_{p,j} \text{ for all } t_u^i t_v^j \in \mathbb{W} t_u^i \neq t_v^j$
- $T t_{j,n+1}^{l} = t m_{i,n+1}$ .
- $T_{0,n+1} = 0.$

The entire model becomes,

$$max.\sum_{i \in T} q_{iY_i}$$

$$\sum_{j \in \delta + (0)} X0jr \sum_{i \in \delta - (n+1)} Xi, n+1, r = 1 \qquad \forall r \in \mathbb{R}$$
(24)

$$\sum_{j\in\delta-(i)} X_{j,i,r} = \sum_{j \in \delta+(i)} X_{i,j,r} \qquad \forall i \in W, r \in \mathbb{R}$$
(25)

$$S(i,1) \le 1 \qquad \forall i \in W$$
 (26)

Constraints (24)–(26) are the common vehicle routing constraints defining the starting and ending location of the tour, flow preservation, and the number of times the UAV can visit the site.

$$S(j+1,1) \le S(j,1)$$
  $\forall i \in T, j \in \{1, ..., n(i) - 1\}$  (27)

$$\sum_{i \in T_i} S(j, qr) \ge q_{iY_i} \qquad \forall i \in \mathcal{T}$$
(28)

Furthermore, Constraint (27) orders the visits: a delivery  $t_{j+1}^i$  cannot me performed whenever delivery  $t_j^i$  has not been made. This constraint, in conjunction with constraint (32), implements the maximum time lag between consecutive deliveries. The sum of capacities of the vehicles performing the deliveries for sites *i*  $\epsilon$  T should cover the demand (Constraint (28).

$$T^{i} - M(1 - X_{ijr}) \leq T^{j} - P_{r} - T_{ij} \qquad \forall (i,j) \in A, i \neq 0, r \in \mathbb{R}$$

$$\tag{29}$$

$$T^{i} - M\left(1 - X_{ijr}\right) \leq T^{j} - T_{ij} \qquad \forall (0,j) \in A, \quad r \in \mathbb{R}$$
(30)

$$T^{i} - S(i, P_{r}) \ge a_{i} \qquad \forall i \in W$$
(31)

$$T^{j+1} - S(j+1, P_r) - T^j \le \gamma \qquad \forall i \in T, j \in \{1, \dots, n(i) - 1\}$$
(32)

$$T^{j+1} \ge T^j + \mathcal{S}(j, P_r) \qquad \forall i \in \mathcal{T}, j \in \{1, \dots, n(i) - 1\}$$

$$(33)$$

Finally, Constraint (33) ensures that visits to the same customer/point do not overlap in time

$$a_i \le T^j \le b_i \qquad \forall \, i \, \epsilon \, V \tag{34}$$

Constraints (29)–(34) enforce the necessary scheduling restrictions. Delivery cannot be made outside the site's time window (Constraints (31) and (33)); travel times need to be accounted for (Constraints (29) and (30)).

$$X_{ijr} \in \{0, 1\} \qquad \forall (i, j) \in A, r \in \mathbb{R}$$
(35)

$$Y_i \in \{0, 1\} \qquad \forall i \in \mathbb{T}$$
(36)

where,  $S(i, \alpha) = \sum r \in \mathbb{R} \sum j \in \delta + (i)a Xijr$  for all i  $\epsilon W$ . Binary variables  $X_{ijr}$  denote whether UAV  $r \epsilon \mathbb{R}$  travels from *i* to *j*, *i*, *j*  $\epsilon V$ . Binary variables  $T^i$  record the time that delivery  $i \epsilon W$  is completed. Additionally,  $T^{n+1}$  records the makespan of the schedule. Finally, Boolean variables  $Y_i$ , denote whether customer  $i \epsilon \mathbb{T}$  is serviced.

#### 4. Results and Discussions

As presented in the method section, the results of GIS, remote sensing, and the PSObased proposed model are presented in this section.

#### 4.1. Monitoring the Burnt Area

NOAA's sensor, VIIRS fire product with 375 m resolution, is used to aggregate the perimeters of the fire with distances of 1 km, 2 km, and 5 km, as shown in Figure 7. The examples show the extent of the land damaged by the Black Summer fire season, where the burnt areas are shown in red colour. The maps show that the eastern part of the state is most affected by the fires. Significant, and more impactful, fire events have hit the cities along the coast, including Sydney, Coffs Harbour, New Castle, and Wollongong. The 1 km aggregates show that the sporadic distribution of the fires is largely spread throughout the state, as evident from Figure 7a. It depicts that each fire event has caused widespread damage. Likewise, the 2 km and 5 km aggregates show the land damages of the respective

distances as shown in Figure 7b,c, respectively. The 5 km aggregates give the best depiction of the burnt area. The areas shown in this map include the southeastern region that connects NSW with the State of Victoria.





**Figure 7.** Examples of aggregates of varying distances using the VIIRS dataset. (**a**) Fire distance 1 km (**b**) Fire distance 2 km (**c**) Fire distance 5 km.

Additionally, the northern parts of the NSW are also heavily destroyed by these fires. Upon visual validation of the output with the Google Earth imagery, it is found that 50% of the national parks of NSW are impacted by the 2019–2020 fire season. This is in line with the NSW report that states a significant impact on NSW vegetation. These fires have resulted in damage of 2.5 million hectares of the state's national parks [68].

#### 4.2. Fire Clustering Patterns and the Significance

Figure 8 shows the level of clustering in the fire data, based on Get-Ord General G statistics generated through ArcGIS. The z-score is based on the randomization of the null hypothesis calculation. The distance method for the clustering analysis is Euclidean distance assessed based on inverse distance. The distance threshold for the state of NSW bushfire events is found to be 148,828.07 m. The higher z-score of 5.90 depicts a less than 1% likelihood that the events' highly clustered pattern could be attributed to random chance. Therefore, these fire events are significantly high clustered along the entire state. Very strong clustering patterns can be visualized in Figure 8 towards the north and south-eastern cities of NSW. Particularly in the southern parts, the clustering trend is quite pronounced mainly because the epicentre of the fire is near the Victoria NSW border region.



**Figure 8.** Graph showing the level of clustering in the fires of Black Summer Fire Season and General G Summary.

The statistical data generated by ArcGIS is presented in Table 5. As the z-score value is positive, and the observed General G index is larger than the expected General G index, high values for the attribute are clustered in the study area. Thus, more evenly distributed fires are experienced in NSW. Therefore, the null hypothesis is accepted as evident from the *p*-value of 0.07, which shows statistical significance when the *p*-value is greater than 0.05. Hence, it is confirmed that all the values linked with each feature are distributed randomly.

Table 5. The statistical values of the data obtained from ArcGIS.

Assessment	Value
Observed General G	0.02
Expected General G	0.005
Variance	0.0
Z-Score	5.9
<i>p</i> -value	0.07

#### 4.3. Monitoring of Bushfire Hotspots

The hotspot analysis, based on the Getis Ord Gi\* statistics, is performed on the aggregated features of 5 km for ease of computation. Figure 9a shows the fire hotspots ranging in five categories: strong cold spots shown in dark blue, cold spots shown in light blue, nonsignificant spots shown in yellow, hotspots shown in orange, and strong hotspots shown in the red colour. Most of the NSW is covered sporadically with strong cold spots, primarily concentrated in the Pilliga Nature Reserve, Wollerni National Park, Yengo National Park, and the coastal regions in the north. Statistically, not-so-significant fire spots could be observed in Blue Mountain's National Park and Morton National Park. Strong hotspots of bushfires are observed in the Deua National Park, situated in the southern part of the state. Some of the clear hotspots could also be observed in the Kosciu Sako National Park.

Similarly, Figure 9b illustrates the graph based on the z-score stats of the Get-Ord Gi<sup>\*</sup> analysis. The graph gives an insight into the types of bushfire hotspots. The number of fires is plotted on the x-axis against their respective z-score values on the y-axis. The blue colour is densely populated across the study area and depicts that more than half of the state is included in the cold spots or strong cold spots. Some random parts of the state bushfires are shown in yellow, depicting that these events are not statistically significant. The relatively low z-score of 2.8–4.1 can be seen in the orange colour, which shows a good clustering of hotspots. On the contrary, the strong z-score values above 4–10.5 depict strong clusters of severe hotspots in the study area. Though these strong hotspots are not so thickly populated across the study area, they impacted the fires throughout the state.

![](_page_20_Figure_5.jpeg)

![](_page_21_Figure_2.jpeg)

**Figure 9.** (a) Delineation of hotspots and cold spots, based on the Getis-Ord Gi\* Statistics, (b) the z-score graph of the hotspots of the study area. Note: blue shows the cold spots, yellow shows the statistically nonsignificant spots, and red shows the strong hotspots.

Figure 10 shows the complete picture of the bushfire hotspots in the state of NSW. The graduated size of circles shows the intensity of the hotspots across the study area. The map shows that the fire events have severely hit the eastern regions of NSW. Minor hotspots of bushfires are densely and randomly dispersed throughout the study area. The noteworthy and more impactful hotspots are shown in a bigger circle in southeastern NSW. These include the Deua National Park, Morton National Park, and Kosciu Sako National Park. Other larger hotspots are in the northeast in the Wollerni and Yengo National Parks. These regions were already at risk of fires, considering the weather conditions of hot maximum temperatures, dry and humid incidents, and the prevalent fire weather situations. Further, it points out that the upcoming fire events will be more frequent and severe than ever recorded before [6]. Thus, the government must act and put measures in place before the next fire event.

![](_page_22_Figure_1.jpeg)

**Figure 10.** (a) Hotspots of bushfires in the 2019–2020 fire season. (b) The eastern side shows strong hotspots for fires and the spatial directional distribution of the fires.

#### 4.4. Regression Analysis of the Black Summer Fires

Most of the northern and central areas of the NSW have observed extremely low precipitation in the year 2019. Some of these locations recorded the driest conditions in history. Lower rainfalls impacted the water resources and the associated firefighting mitigating measures [69]. By the start of August, almost the entire NSW was stricken by severe drought (55%), observing drought conditions (23%), and experiencing extreme droughts (17%). The initial 'Section 4.4 emergency' was declared on 10 August 2019 [70]. Additionally, adequate soil moisture deficits and prevalent winds facilitated the considerable frequency of fire events [71]. A total area of 5,595,739 hectares was burnt, destroying 2475 houses, and causing 25 casualties by 10,520 fire incidents in NSW, as shown in Figure 11.

![](_page_23_Figure_4.jpeg)

#### 2000 J-01 J-02 J-03 J-04 J-05 J-06 J-07 J-08 J-09 J-10 J-11 J-12 J-13 J-14 J-15 J-16 J-17 J-18 J-19 J-20

**Figure 11.** The numbers of fire events in each season in NSW since 2001. (Note J-01 means January 2001, J-02 means January 2002 and so on. The x-axis shows the month and year, whereas the y-axis shows the number of fires.).

These fires of NSW made a record of burning more area than any other fire season in the past two decades, as seen in Figure 11, that the authors compiled based on the data available online. Figure 11 shows data for January of each odd calendar year starting from 2001 till 2019, where the number of fires is plotted. The 2019 fires have been the worst disaster in terms of the number of fires and areas burnt. The year 2012 had the least burnt area and lower numbers of fire, followed by 2008 and 2004.

Figure 12 shows the data for houses burnt and the fatalities of various fires in NSW since 2001. All the data is plotted for the reports of January of the particular year. The authors compiled the data based on online information and reports of the parliament of Australia and NSW RFS. The Black Summer fire season has been an extraordinary disaster where the burnt area, fatalities, and damaged houses are more significant than the previous years. Before the 2019 season, the notable instances are 2001 and 2002 fires, with 250 houses damaged and two fatalities each. The 2013–2014 fires resulted in the loss of three lives and damages to 350 houses. Similarly, 2013 and 2017 also resulted in the loss of two lives and damages to approx. 200 properties.

![](_page_24_Figure_1.jpeg)

**Figure 12.** Bushfire impacts on houses and fatalities in each fire season in NSW since 2001. (Note: J means January, A means April, O means October, and D means December. The numbers after J, A, O, D refers to the year of data points). (a): Number of houses burnt in various months since 2001 (the x axis shows the months and years, whereas the y-axis shows the number of burnt houses). (b): Number of lives lost each year since 2001 (the x axis shows the months and years, whereas the y-axis shows the number of fatalities).

In Table 6, before the Black Summer fires, the trend of the burnt area showed a negative slope, transitioning into a positive with the 2019–2020 fires, signifying a *p*-value of 0.019. The frequency of fires has been decreasing until 2012 that started to increase post-2012. It showed a positive trend for the dataset with a greater slope for 2001–2020. The data analysis shows a positive linear relationship between the fire events and the burnt area that is found to be statistically significant with a *p*-value of 0.59. A regression line for the house damaged over the years shows a positive slope. It shows a statistically significant *p*-value of 0.12. Due to the 2019–2020 fire season data, a statistically significant output with a *p*-value of 0.54 is obtained. Fatalities are estimated to be about 1% of the houses damaged. This dataset shows an error of 0.36 for the fatalities. The results are similar to an extensive study performed by Filkov et al. [72], where the researchers explored the impact of the recent fires on the houses and lives lost.

Table 6. Regression Analysis for 2001–2020 fire seasons in New South Wales.

Impact Factors	Standard Error	p	r	R2
Fire Season (y) vs. Burnt Area (x)	1,288,108	0.019	0.59	0.35
Fire Season (y) vs. Houses Loss (x)	52,737	0.12	0.39	0.15
Fire Season(y) vs. Life Loss (x)	5.30	0.11	0.40	0.16
Burnt Area(y) vs. Fire Season (x)	1639	2.75 × 10 <sup>-19</sup>	0.59	0.31
Life Loss (y) vs. Houses Loss (x)	0.36	0.54	0.99	0.99

#### 4.5. UAV Routing Results

As discussed in the Method Section, the UAVs' paths were optimized using the PSO algorithm to have the shortest possible distance for monitoring the fire events. For doing this, different cases are considered in the target area of NSW Australia. Varying iterations, number of UAVs, computation time taken by the control unit, and the best (shortest) travel distance for UAVs are presented in Table 7. The iterations include 50, 100, 200, 300, 400, and 500, whereas the number of UAVs is varied between 20, 40, 60, 80, and 100 for each iteration. From Table 7, in most of the cases, 20 UAVs give the best results for computation speeds, and 100 UAVs give the best transportation distance to be covered (shortest distance). The computation time ranges from 14.64 s for 20 UAVs and 50 iterations to 137.37 s for the same number of UAVs with 500 iterations. On average, the optimized distance is around 12.81 km of area, which is a considerable distance, considering the limited battery operating time of the UAV. This shows that the more UAVs there are in the swarm, the better the results will be, as the UAVs can communicate with more UAVs in the swarm and share the workload more efficiently. Thus, depending on the area to be covered, the UAVs in the swarm should be increased. The values are generated through the MATLAB code for PSO to optimize the UAV routes.

No. of Iteration (I)	Number of UAVs ( <i>n</i> )	Elapsed Time(s)	Best Transportation Distance (km)
	20	14.64	12.89
	40	16.24	12.97
50	60	18.49	13.24
	80	20.34	13.18
	100	22.07	12.81
	20	28.36	13.08
	40	32.65	12.85
100	60	35.65	12.92
	80	40.03	13.16
	100	43.20	12.80
200	20	57.37	12.82

Table 7. Test cases with the number of iterations, UAVs, elapsed computation time, and optimized distance.

	40	63.72	12.80	
	60	71.12	12.86	
	80	78.85	12.91	
	100	85.14	12.80	
	20	85.37	13.24	
	40	96.17	13.33	
300	60	106.80	13.34	
	80	116.07	13.21	
	100	128.06	12.82	
	20	110.97	13.09	
	40	123.71	13.01	
400	60	139.32	12.86	
	80	152.12	13.19	
	100	167.84	12.80	
	20	137.37	13.20	
	40	153.32	13.21	
500	60	173.50	13.01	
	80	191.89	12.85	
	100	204.85	12.82	

There is a trade-off between computational time and transportation cost. Moreover, considering the number of iterations to be 200 and 400 for 100 UAVs, the best transportation distance is the same (12.80 km); however, the elapsed time is 85.14 and 167.84 s. This shows that the time can vary even with the same cost; hence, there is no universal rule for selecting a definite number of UAVs or inferring that the maximum number of UAVs will give minimum cost in all cases.

Figure 13 shows the iteration results for all test cases involving 100 UAVs. The optimal UAV path for detecting bushfires is simulated in the MATLAB environment. For this purpose, a PSO algorithm is designed based on certain parameters such as iteration, population size, inertial weight, damping ratio, as well as personal and global learning coefficient to calculate the best transportation distance. Figure 13a,c,e,g,i,k illustrates a 2D scenario of UAV routing problem from starting point to destination. Blue circles are the obstacles in the UAV trajectory. The best transportation distance and elapsed time are calculated based on the UAV hovering from the start to the destination, where the effect zone (bushfire) needs to be monitored. Three handle points, indicated in light red circles, are considered to smoothen the path from start to destination. The x and y-axis represent the lengths and widths of the plots or locations (P1 to P10). Figure 13b,d,f,h,j,l shows the best transportation distance, concerning number of iterations. In all the cases, the case study area's optimized distance with 100 UAVs is 12.81 km, on average. When the iterations are increased from a specific point, the distance follows a straight path. In the case of the lower number of UAVs, the optimized distance is increased, which means lower productivity and more battery losses. Thus, it is advised to use more UAVs in the swarm for better results and less travel distances for swift disaster response.

![](_page_27_Figure_2.jpeg)

![](_page_28_Figure_2.jpeg)

**Figure 13.** The 2D trajectory of the UAVs routing and the shortest transportation distance for test cases with 100 UAVs. (a) 2D Scenario for 50 iterations and 100 UAVs (b) Transportation distance for 50 iterations and 100 UAVs (c) 2D Scenario for 100 iterations and 100 UAVs (d) Transportation distance for 100 iterations and 100 UAVs (e) 2D Scenario for 200 iterations and 100 UAVs (f) Transportation distance for 200 iterations and 100 UAVs (g) 2D Scenario for 300 iterations and 100 UAVs (h) Transportation distance for 200 iterations and 100 UAVs (g) 2D Scenario for 300 iterations and 100 UAVs (i) 2D Scenario for 300 iterations and 100 UAVs (i) 2D Scenario for 300 iterations and 100 UAVs (i) 2D Scenario for 400 iterations and 100 UAVs (j)

Transportation distance for 400 iterations and 100 UAVs (**k**) 2D Scenario for 500 iterations and 100 UAVs (**l**) Transportation distance for 500 iterations and 100 UAVs.

Figure 14 shows a swarm of UAVs deployed at a test zone at NSW with their starting points, destination point, and various possible routes. Using PSO, the Route 1 value is 12.89 km with a computation time of 14.63 s. Route 2 takes 18.49 s and gives an optimized path of 13.24 km, whereas Route 3 has values of 12.80 km for the optimized path and takes 22.69 s to be computed. Depending upon the scenario, if computation time is not the highest priority, R3 has the best results for the optimized path. However, if computation speed is the concern, Route 1 has the best values with just 14.63 s of computation time. Thus, the trade-off between computation speeds and optimized routes may be input into the control units through artificial intelligence or human presence. This decision can be taken in real-time at the control unit or head office through remote administration.

![](_page_29_Figure_3.jpeg)

Figure 14. PSO-based optimized routes and computation speeds for UAVs in a test zone at NSW.

Accordingly, the NSW and the rest of Australia can be covered through UAV swarms whose paths are optimized through PSO algorithms to tackle any bushfire disaster. The proposed system can be adopted by the NSW RFS to plan for upcoming fire seasons actively. The system, if adopted, can help save lives, reduce the bushfire impacts on properties and livestock, and save many species of wildlife from bushfire disasters. With the growing availability of UAVs, the proposed system will cost way less than the post-disaster rehabilitation and repairs. Vigilant and swift policy making is required in this context to help mitigate the harmful effects of bushfires disasters by adopting the proposed system.

The area covered depends on the availability of UAVs and the technology level used in the UAVs. As far as the costs associated with the increased number of UAVs are considered, it can be estimated using the area coverage path planning, which depends on finding the route that covers every point within the target area of interest [73]. Particularly, for our system, the costs of using an increased number of UAVs will be greatly dependent on the area coverage path planning for our area of interest.

In various disastrous situations, many UAVs can be effectively deployed, as the utility of UAVs is highly implicated and appreciated in such future situations. The UAVs can play promising roles in various relief efforts in natural disaster situations such as bushfires, storms, and earthquakes. Notably, UAVs can perform very crucial jobs as soon as disasters emerge. These tasks include identifying people in emergencies who require urgent help. Greater evidence supports the fact that UAVs display several advantages over traditional searching and rescuing with a considerably higher speed. UAVs can perform additional roles, such as delivering the rescue ropes and life jackets in dangerous areas where ground-based rescue efforts are practically impossible and very difficult. These UAVs have an inherent ability to assess the damage caused to infrastructures such as roads, buildings, tunnels, bridges, etc. Different types of models can be effectively coupled with these systems. Some of the most prominent examples are the traditional continuum model, expand contract model, and disaster crunch model. The traditional continuum model is based on sequential stages, focusing on the activities associated with pre-disaster and post-disaster events. Our proposed systems can be effectively used if integrated with such types of models. The coupling of our systems to these models will speed up and enhance the projection, which will help attain an improved response from the governing bodies. If the governing bodies for disaster management extend their support, UAVs can be effectively used in natural disasters such as floods, bushfires, etc.

During the post-disaster assessment, the battery life of the UAV plays a significant role in hovering time. In a recent study by Fotouhi et al. [74], the authors implemented a control strategy for confined Phantom 4 mobility using a DJI software development kit (SDK). Considerably, the speed of the UAV is directly proportional to the power utilized. The pertinent results shows that the power utilization abruptly reaches 167 W, as the UAV speed peaks to 10 m/s.

UAV battery consumption is always an important issue to address, keeping the sustainability of technology in mind. A solution for the UAV battery energy consumption is proposed in Selim and Kamal [75] using UAV Base Station (UAV-BS) and Powering Drone (PD). The PD provides the necessary charging for the hovering UAV-BS to make it more efficient for monitoring the affected area without going back to the depot and providing the optimum results. In the relevant study, during the initial timing block, from 0 to 1, the UAV-BS are initialized from 200 kJ capacity and consume the maximum energy to reach the allocated spot.

## 5. Conclusions

The current study investigated the devastating 2019–2020 Black Summer fires occurring in NSW Australia. Using the case study of the NSW region of Australia, GIS and remote sensing analyses were conducted to map the burnt areas of NSW. The results highlight that 50% of the national parks of NSW were impacted by the 2019–2020 fire season, resulting in damage to 2.5 million hectares of the state's national parks. The fire clustering patterns indicated that these events are significantly, highly clustered in the entire state, where very strong clustering patterns can be visualized towards the north and southeastern cities of NSW. The clustering trend is quite pronounced on the southern side of NSW, where it shares the border with Victoria.

Similarly, the hotspot mapping shows that strong hotpots of bushfires are in the Deua National Park, situated in the southern part of the state, and the Kosciu Sako National Park. Other larger hotspots are in the northeast in the Wollerni and Yengo National Parks, which had been declared to be at risk by the government due to weather conditions of hot maximum temperatures, dry and humid incidents, and the prevalent fire weather situations. The government must act and enact measures before the next fire event; otherwise, the data trends show that upcoming fires could be more devastating than the Black Summer.

A UAV-based bushfire monitoring system is proposed in the current study to monitor the bushfires in NSW and instigate a swift response plan to minimize the losses. The paths of the UAVs are planned and optimized using the PSO algorithm for avoiding barriers in the path and covering the maximum area in the shortest possible time. The test results with 50, 100, 200, 300, 400, and 500 iterations and the number of UAVs varying between 20, 40, 60, 80, and 100 for each iteration show that 20 UAVs give the best results for computation speeds, and 100 UAVs give the best transportation distance to be covered. Thus, the more UAVs are there in the swarm, the better will be the results in most cases. These UAVs can communicate with more UAVs in the swarm and share the workload more efficiently if the number is higher. Thus, it is proposed to increase UAVs if more area is to be covered and monitored.

The current study is limited to test cases without submission to field tests due to the non-fire seasons. In the future, it can be tested in the field, and real-time results can be assessed to test the in-field validity and performance of the proposed method. Nevertheless, it is a first step towards addressing a key issue of bushfire disasters in the Australian context that other countries in the world can adopt. For Australia and NSW, the RFS can adopt the proposed system and have the UAV swarm ready before the next fire season to instantly map, assess, and mitigate bushfire disasters. Further, it is recommended to develop a shared, integrated platform for diverse data sources, intelligence, and information sharing across government organizations where useful data can be shared. New wildfire risk assessments should be conducted with high-resolution mapping technologies to assess the current state of wildlife and help place protection measures in place. The scientific understanding of "megafires" should be enhanced through retrospective analysis and fire behaviour models, and associated inputs for real-time prediction should be investigated. These, when achieved in their true essence, will help lay the foundation of enhanced environmental sustainability for industry 5.0.

Author Contributions: Conceptualization, F.U. and H.S.M.; methodology, F.U., S.I.K., H.S.M., Z.Q. and S.Q.; software, F.U., S.I.K., H.S.M., Z.Q. and S.Q.; validation, F.U. and H.S.M.; formal analysis, F.U., S.I.K. and H.S.M.; investigation, F.U., S.I.K. and H.S.M.; resources, F.U. and H.S.M.; data curation, F.U., S.I.K., H.S.M., Z.Q. and S.Q.; writing—original draft preparation, F.U., SI.K. and H.S.M.; writing—F.U., S.I.K., H.S.M., Z.Q. and S.Q.; visualization, F.U., S.I.K., H.S.M., Z.Q. and S.Q.; visualization, F.U., S.I.K., H.S.M., Z.Q. and S.Q.; visualization, F.U., S.I.K., H.S.M., Z.Q. and S.Q.; supervision, F.U.; project administration, F.U. and H.S.M.; funding acquisition, F.U. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

**Data Availability Statement:** The data used in the study is available with the authors and can be shared upon reasonable requests.

**Availability of Codes:** The MATLAB codes used in the study are available with the authors and can be shared upon reasonable requests. The PSO optimization algorithm used in the Bushfire Application is inspired by "Seyedali Mirjalili (2021). Simulation of particles in Particle Swarm Optimization (https://www.mathworks.com/matlabcentral/fileexchange/69027-simulation-of-particles-in-particle-swarm-optimization), MATLAB Central File Exchange. Retrieved 9 August 2021.

Conflicts of Interest: The authors declare no conflict of interest.

## References

- Atif, S.; Umar, M.; Ullah, F. Investigating the flood damages in Lower Indus Basin since 2000: Spatiotemporal analyses of the major flood events. *Nat. Hazards* 2021, 108, 2357–2383.
- Halofsky, J.E.; Peterson, D.L.; Harvey, B.J. Changing wildfire, changing forests: The effects of climate change on fire regimes and vegetation in the Pacific Northwest, USA. *Fire Ecol.* 2020, *16*, 4.
- 3. Parente, J.; Pereira, M.; Amraoui, M.; Fischer, E.M. Heat waves in Portugal: Current regime, changes in future climate and impacts on extreme wildfires. *Sci. Total Environ.* **2018**, *631*, 534–549.
- Vilà-Vilardell, L.; Keeton, W.S.; Thom, D.; Gyeltshen, C.; Tshering, K.; Gratzer, G. Climate change effects on wildfire hazards in the wildland-urban-interface–Blue pine forests of Bhutan. *For. Ecol. Manag.* 2020, 461, 117927.

- Munawar, H.S.; Ullah, F.; Khan, S.I.; Qadir, Z.; Qayyum, S. UAV Assisted Spatiotemporal Analysis and Management of Bushfires: A Case Study of the 2020 Victorian Bushfires. *Fire* 2021, *4*, 40.
- 6. Di Virgilio, G.; Evans, J.P.; Blake, S.A.; Armstrong, M.; Dowdy, A.J.; Sharples, J.; McRae, R. Climate change increases the potential for extreme wildfires. *Geophys. Res. Lett.* **2019**, *46*, 8517–8526.
- Matthews, S.; Sullivan, A.L.; Watson, P.; Williams, R.J. Climate change, fuel and fire behaviour in a eucalypt forest. *Glob. Chang. Biol.* 2012, 18, 3212–3223.
- 8. Doerr, S.H.; Santín, C. Global trends in wildfire and its impacts: Perceptions versus realities in a changing world. *Philos. Trans. R. Soc. B Biol. Sci.* **2016**, *371*, 20150345.
- Konovalov, I.; Beekmann, M.; Kuznetsova, I.; Yurova, A.; Zvyagintsev, A. Atmospheric impacts of the 2010 Russian wildfires: Integrating modelling and measurements of an extreme air pollution episode in the Moscow region. *Atmos. Chem. Phys.* 2011, 11, 10031.
- 10. Gilbert, N. Russia counts environmental cost of wildfires. Nature 2010, 12, doi:10.1038/news.2010.404.
- 11. Read, P.; Denniss, R. With Costs Approaching \$100 Billion, the Fires Are Australia's Costliest Natural Disaster. Available online: https://theconversation.com/with-costs-approaching-100-billion-the-fires-are-australias-costliest-natural-disaster-129433#:~:text=With%20costs%20approaching%20%24100%20billion%2C%20the%20fires%20are%20Australia's%20costliest%2 0natural%20disaster (accessed on 31 October 2020).
- 12 Oliver, The Impact the Australian Bushfires Worldwide. S. of on Investors Available online: https://www.ampcapital.com/au/en/capital-edition/edition-5/the-impact-of-the-australian-bushfires-on-investors-worldwide (accessed on 31 October 2020).
- Parliment, A. 2019–20 Australian Bushfires—Frequently Asked Questions: A Quick Guide. Available online: https://parlinfo.aph.gov.au/parlInfo/download/library/prspub/7234762/upload\_binary/7234762.pdf (accessed on 31 October 2020).
- 14. Gordon, C. Remembering the 1965 Chatsbury-Bungonia Fire; Goulburn Post: Goulburn, Australia, 2015.
- Parliament, N. Issues Backgrounder. Bushfires in NSW: Timelines and Key Sources. Number 6/June 2014. NSW Parliamentary Research Service. Available online: https://www.parliament.nsw.gov.au/researchpapers/Documents/bushfires-in-nswtimelines-and-key-sources/Bushfires%20in%20NSW%20-%20timelines%20and%20key%20sources.pdf (accessed on 1 November 2020).
- Ministry for Police & Emergency Service. 1979–1980, Sydney and Region Bushfire. Available online: https://web.archive.org/web/20131023061443/http://www.emergency.nsw.gov.au/content.php/602.html (accessed on 1 November 2020).
- 17. EMA. Bushfires—Get the Facts. Available online: https://web.archive.org/web/20130116110828/http://www.ema.gov.au/www/ema/schools.nsf/Page/Get\_The\_FactsBushfires (accessed on 1 November 2020).
- 18. NSW Gov. Christmas 2001/2002 Bushfires; NSW, F.R., Ed.; NSW Gov: Sydney, Australia, 2007.
- 19. SMH. Bushfire threat eases in NSW. The Sydney Morning Herald, 4 January 2006.
- 20. Knox, J.; Mathew, S. Can do attitude at Wandoo fire 6-19 February 2006 Wagga Section 44. *BushFire Bull.-J. NSW Rural. Fire Serv.* **2006**, *28*, 17.
- 21. Kennedy, L.; Braithwaite, D.; Tadros, E. Man dies as early bushfire season grips NSW. The Age, 22 November 2006.
- 22. van de Wetering, J. A timeline of the Coonabarabran Bushfires. Available online: https://www.abc.net.au/local/stories/2013/02/13/3689707.htm (accessed on 1 November 2020).
- 23. Madden, J. Firestorm destroys NSW Communities as hundreds of homes could be lost. *The Australian*, 18 October 2013.
- SBS. 35 Homes Razed in NSW Blazes. Available online: https://www.sbs.com.au/news/nsw-residents-tell-of-terrifying-bushfire (accessed on 16 May 2021).
- RFS, N. Final Report of the NSW Bushfire Inquiry. Available online: https://apo.org.au/sites/default/files/resource-files/2020-07/apo-nid307786.pdf (accessed on 1 November 2020).
- 26. Euchi, J. Do drones have a realistic place in a pandemic fight for delivering medical supplies in healthcare systems problems. *Chin. J. Aeronaut.* **2021**, *34*, 182–190.
- 27. Tsouros, D.C.; Bibi, S.; Sarigiannidis, P.G. A review on UAV-based applications for precision agriculture. *Information* **2019**, *10*, 349.
- Ullah, F.; Sepasgozar, P.S.; Ali, T.H. Real estate stakeholders technology acceptance model (RESTAM): User-focused big9 disruptive technologies for smart real estate management. In Proceedings of the 2nd International Conference on Sustainable Development in Civil Engineering (ICSDC 2019), Jamshoro, Pakistan, 5–7 December 2019.
- Felli, F.; Liu, C.; Ullah, F.; Sepasgozar, S. Implementation of 360 videos and mobile laser measurement technologies for immersive visualisation of real estate & properties. In Proceedings of the 42nd AUBEA Conference, Singapore, 26–28 September 2018.
- Ullah, F.; Sepasgozar, S. A study of information technology adoption for real-estate management: A system dynamic model. *Innov. Prod. Constr.* 2019, 469–486, doi:10.1142/9789813272491\_0027.
- 31. Ullah, F.; Sepasgozar, S.M. Key Factors Influencing Purchase or Rent Decisions in Smart Real Estate Investments: A System Dynamics Approach using Online Forum Thread Data. *Sustainability* **2020**, *12*, 4382.

- 32. Ullah, F.; Sepasgozar, S.M.; Wang, C. A systematic review of smart real estate technology: Drivers of, and barriers to, the use of digital disruptive technologies and online platforms. *Sustainability* **2018**, *10*, 3142.
- Hildmann, H.; Kovacs, E. Using Unmanned Aerial Vehicles (UAVs) as Mobile Sensing Platforms (MSPs) for Disaster Response, Civil Security and Public Safety. Drones 2019, 3, 59.
- Munawar, H.S.; Hammad, A.; Ullah, F.; Ali, T.H. After the flood: A novel application of image processing and machine learning for post-flood disaster management. In Proceedings of the 2nd International Conference on Sustainable Development in Civil Engineering (ICSDC 2019), Jamshoro, Pakistan, 5–7 December 2019.
- 35. Munawar, H.S.; Qayyum, S.; Ullah, F.; Sepasgozar, S. Big Data and Its Applications in Smart Real Estate and the Disaster Management Life Cycle: A Systematic Analysis. *Big Data Cogn. Comput.* **2020**, *4*, 4.
- Munawar, H.S.; Ullah, F.; Qayyum, S.; Khan, S.I.; Mojtahedi, M. UAVs in Disaster Management: Application of Integrated Aerial Imagery and Convolutional Neural Network for Flood Detection. *Sustainability* 2021, 13, 7547.
- 37. Greenwood, F.; Nelson, E.L.; Greenough, P.G. Flying into the hurricane: A case study of UAV use in damage assessment during the 2017 hurricanes in Texas and Florida. *PLoS ONE* **2020**, *15*, e0227808.
- Wang, N. "We Live on Hope...": Ethical Considerations of Humanitarian Use of Drones in Post-Disaster Nepal. *IEEE Technol.* Soc. Mag. 2020, 39, 76–85.
- Van Der Werf, G.R.; Randerson, J.T.; Giglio, L.; van Leeuwen, T.T.; Chen, Y.; Rogers, B.M.; Mu, M.; van Marle, M.J.; Morton, D.C.; Collatz, G.J. Global fire emissions estimates during 1997–2016. *Earth Syst. Sci. Data* 2017, *9*, 697–720.
- Waigl, C.F.; Stuefer, M.; Prakash, A.; Ichoku, C. Detecting high and low-intensity fires in Alaska using VIIRS I-band data: An improved operational approach for high latitudes. *Remote Sens. Environ.* 2017, 199, 389–400.
- 41. Hantson, S.; Padilla, M.; Corti, D.; Chuvieco, E. Strengths and weaknesses of MODIS hotspots to characterize global fire occurrence. *Remote Sens. Environ.* **2013**, *131*, 152–159.
- 42. Chuvieco, E.; Aguado, I.; Jurdao, S.; Pettinari, M.L.; Yebra, M.; Salas, J.; Hantson, S.; de la Riva, J.; Ibarra, P.; Rodrigues, M. Integrating geospatial information into fire risk assessment. *Int. J. Wildland Fire* **2014**, *23*, 606–619.
- 43. Giglio, L.; Boschetti, L.; Roy, D.P.; Humber, M.L.; Justice, C.O. The Collection 6 MODIS burned area mapping algorithm and product. *Remote Sens. Environ.* 2018, 217, 72–85.
- 44. Giglio, L.; Schroeder, W.; Justice, C.O. The collection 6 MODIS active fire detection algorithm and fire products. *Remote Sens. Environ.* **2016**, *178*, 31–41.
- 45. Hart, T.; Zandbergen, P. Kernel density estimation and hotspot mapping. *Polic. Int. J. Police Strateg. Manag.* 2014, 37, doi:10.1108/PIJPSM-04-2013-0039.
- 46. Getis, A.; Ord, J. The analysis of spatial association by use of distance statistics, geographycal analysis. In *Perspectives on Spatial Data Analysis*; Springer: Berlin/Heidelberg, Germany, 1992.
- 47. Ord, J.K.; Getis, A. Local spatial autocorrelation statistics: Distributional issues and an application. *Geogr. Anal.* **1995**, 27, 286–306.
- Graml, R.; Wigley, G. Bushfire hotspot detection through uninhabited aerial vehicles and reconfigurable computing. In Proceedings of the 2008 IEEE Aerospace Conference, Big Sky, MT, USA, 1–8 March 2008.
- 49. Christensen, B. Use of UAV or remotely piloted aircraft and forward-looking infrared in forest, rural and wildland fire management: Evaluation using simple economic analysis. N. Z. J. For. Sci. 2015, 45, 16.
- Lim, Y.; Samreeloy, T.; Chantaraviwat, C.; Ezer, N.; Gardi, A.; Sabatini, R. Cognitive human-machine interfaces and interactions for multi-UAV operations. In Proceedings of the 18th Australian International Aerospace Congress (AIAC18), Melbourne, Australia, 24–28 February 2019.
- Brunori, E.; Maesano, M.; Moresi, F.V.; Antolini, A.; Bellincontro, A.; Forniti, R.; Biasi, R.; Mencarelli, F. Using UAV-based Remote Sensing to Assess Grapevine Canopy Damages Due to Fire Smoke. J. Sci. Food Agric. 2020, 100, 4531–4539.
- 52. Gissing, A.; Bibby, N. Readiness for the next major bushfire emergency. Aust. J. Emerg. Manag. 2020, 35, 13.
- Ullah, F.; Al-Turjman, F.; Qayyum, S.; Inam, H.; Imran, M. Advertising through UAVs: Optimized path system for delivering smart real-estate advertisement materials. *Int. J. Intell. Syst.* 2021, *36*, 3429–3463.
- Qadir, Z.; Ullah, F.; Munawar, H.S.; Al-Turjman, F. Addressing disasters in smart cities through UAVs path planning and 5G communications: A systematic review. *Comput. Commun.* 2021, 168, 114–135.
- 55. Adamu, P.I.; Jegede, J.T.; Okagbue, H.I.; Oguntunde, P.E. Shortest path planning algorithm–A Particle Swarm Optimization (PSO) approach. In Proceedings of the World Congress on Engineering, London, UK, 4–6 July 2018.
- Ranaweera, D.M.; Hemapala, K.U.; Buddhika, A.; Jayasekara, P. A shortest path planning algorithm for PSO base firefighting robots. In Proceedings of the 2018 Fourth International Conference on Advances in Electrical, Electronics, Information, Communication and Bio-informatics (AEEICB), Chennai, India, 27–28 February 2018.
- 57. Song, B.; Wang, Z.; Zou, L. An improved PSO algorithm for smooth path planning of mobile robots using continuous highdegree Bezier curve. *Appl. Soft Comput.* **2021**, *100*, 106960.
- Dziwiński, P.; Bartczuk, Ł. A new hybrid particle swarm optimization and genetic algorithm method controlled by fuzzy logic. IEEE Trans. Fuzzy Syst. 2019, 28, 1140–1154.
- 59. Wu, D.; Jiang, N.; Du, W.; Tang, K.; Cao, X. Particle swarm optimization with moving particles on scale-free networks. *IEEE Trans. Netw. Sci. Eng.* **2018**, *7*, 497–506.
- 60. Tian, S.; Li, Y.; Kang, Y.; Xia, J. Multi-robot path planning in wireless sensor networks based on jump mechanism PSO and safety gap obstacle avoidance. *Future Gener. Comput. Syst.* **2021**, *118*, 37–47.

- 61. Hang, P.; Huang, S.; Chen, X.; Tan, K.K. Path planning of collision avoidance for unmanned ground vehicles: A nonlinear model predictive control approach. *Proc. Inst. Mech. Eng. Part I J. Syst. Control Eng.* **2021**, 235, 222–236.
- 62. Liu, G.; Chen, X.; Zhou, R.; Xu, S.; Chen, Y.-C.; Chen, G. Social learning discrete Particle Swarm Optimization based two-stage X-routing for IC design under Intelligent Edge Computing architecture. *Appl. Soft Comput.* **2021**, *104*, 107215.
- 63. Pluhacek, M.; Senkerik, R.; Viktorin, A.; Kadavy, T. Exploring the shortest path in PSO communication network. In Proceedings of the 2017 IEEE Symposium Series on Computational Intelligence (SSCI), Honolulu, HI, USA, 27 November–1 December 2017.
- 64. Grayson, B.; Penna, N.T.; Mills, J.P.; Grant, D.S. GPS precise point positioning for UAV photogrammetry. *Photogramm. Rec.* 2018, 33, 427–447.
- 65. Hemmelder, S.; Marra, W.; Markies, H.; de Jong, S.M. Monitoring river morphology & bank erosion using UAV imagery–A case study of the river Buëch, Hautes-Alpes, France. *Int. J. Appl. Earth Obs. Geoinf.* **2018**, *73*, 428–437.
- Mirjalili, S.; Wang, G.-G.; Coelho, L.D.S. Binary optimization using hybrid particle swarm optimization and gravitational search algorithm. *Neural Comput. Appl.* 2014, 25, 1423–1435.
- 67. Mirjalili, S.; Lewis, A.; Sadiq, A.S. Autonomous particles groups for particle swarm optimization. *Arab. J. Sci. Eng.* **2014**, *39*, 4683–4697.
- 68. NSW National Parks and Wildlife Service. Managing Fire in Parks and Reserves. Available online: https://www.nationalparks.nsw.gov.au/about-npws/managing-fire-in-parks (accessed on 1 November 2020).
- 69. BNHCRC. Australian Seasonal Bushfire Outlook: August 2019 Hazard Notes. Available online: https://www.bnhcrc.com.au/hazardnotes/63 (accessed on 12 June 2021).
- 70. Austlii. Rural Fires Act 1997-As at 23 September 2020-Act 65 of 1997; Austlii: Ultimo, Australia, 2020.
- Nolan, R.H.; Boer, M.M.; Collins, L.; de Dios, V.R.; Clarke, H.; Jenkins, M.; Kenny, B.; Bradstock, R.A. Causes and consequences of eastern Australia's 2019–20 season of mega-fires. *Glob. Chang. Biol.* 2020, 26, 1039–1041.
- 72. Filkov, A.I.; Ngo, T.; Matthews, S.; Telfer, S.; Penman, T.D. Impact of Australia's catastrophic 2019/20 bushfire season on communities and environment. Retrospective analysis and current trends. J. Saf. Sci. Resil. 2020, 1, 44–56.
- 73. Cabreira, T.M.; Brisolara, L.B.; Ferreira, P.R., Jr. Survey on coverage path planning with unmanned aerial vehicles. *Drones* **2019**, 3, 4.
- 74. Fotouhi, A.; Ding, M.; Hassan, M. DroneCells: Improving spectral efficiency using drone-mounted flying base stations. *J. Netw. Comput. Appl.* **2021**, *174*, 102895.
- Selim, M.Y.; Kamal, A.E. Post-disaster 4G/5G network rehabilitation using drones: Solving battery and backhaul issues. In Proceedings of the 2018 IEEE Globecom Workshops (GC Wkshps), Abu Dhabi, United Arab, 9–13 December 2018.